Title: Magnetic resonance imaging and histopathological findings in a patient with subcutaneous paragonimiasis: a case report

Authors:

Makoto Kodama (mkodpth1@tmd.ac.jp)
Mayumi Akaki (mayumi_akaki@med.miyazaki-u.ac.jp)
Hiroyuki Tanaka (hiroyuki_tanaka@med.miyazaki-u.ac.jp)
Haruhiko Maruyama (hikomaru@med.miyazaki-u.ac.jp)
Eiji Nagayasu (enagayas@med.miyazaki-u.ac.jp)
Tetsuhiro Yokouchi (skin-plast@yokouchiclinic.jp)
Yasuji Arimura (yasuji_arimura@med.miyazaki-u.ac.jp)
Hiroaki Kataoka (mejina@med.miyazaki-u.ac.jp)

Version: 2
Date: 18 July 2014

Author's response to reviews: see over
Dear Ms. Vidal:

Thank you very much for your e-mail dated July 14th, 2014. We hereby submit a modified version of our manuscript, MS: 5995776571354771, entitled “Magnetic resonance imaging and histopathological findings in a patient with subcutaneous paragonimiasis: a case report” written by Kodama, et al., as a case report to the Journal of Medical Case Reports. Along with the manuscript, we also submit the Word file of the CARE checklist as required by the Journal.

Following your instructions, we have changed the description of the patient’s gender to “female” and added a list of abbreviations used in the manuscript after the Consent section. Additionally, in the Discussion section (page 8), rationale for our final diagnosis has been more clearly stated as recommended by the CARE Guidelines. We hope you find our manuscript suitable for the Journal style and look forward to hearing from you.

Sincerely,

Hiroaki Kataoka, M.D., Ph.D.
Professor
Section of Oncopathology and Regenerative Biology,
Department of Pathology, Faculty of Medicine, University of Miyazaki